Author’s response to reviews

Title: Emergency inguinal hernioplasties in a Tertiary Public Hospital in Athens Greece, during the Economic Crisis

Authors:

Ioannis Karavokyros (iokaravokyros@msn.com)
George Kirkilessis (geokirkilessis@yahoo.gr)
Demetrios Schizas (schizasd@otenet.gr)
Georgios Chelidonis (gchelido@yahoo.gr)
Emmanouil Pikoulis (mpikoul@med.uoa.gr)
John Griniatsos (johngmniatsos@yahoo.com)

Version: 4 Date: 18 Jan 2019

Author’s response to reviews:

As it can be seen in page 4 Line 1 ‘Introduction’ was changed to ‘Background’, and in Line 25 ‘Materials and methods’ to ‘Methods’.

As it can be seen in page 14 Line 1 to page 16 Line 1, all subsections within the Declarations section have headings and the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number have been added.

The Response to reviewers/Cover Letter have been removed from the submission’s supplementary files.

The manuscript and the relevant tables have been uploaded as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours with the exception of hyperlinks or email addresses.